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Live Time: 95.9% good physics data

POT Fraction: 80.1% good physics data

€ Target profile monitor was apparently out
of the beam between Sunday Oct 2, night
and 10am Monday Oct 3

@ For this period, we automatically record
bad beam, must manually intervene to
bypass beam data quality cuts

€ Beam data group is aware and will fix this

€ Good data at POT fraction of 96%

€ Good physics data will be at ~96% as well
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Status Report

Live Time: 96.7% good physics data
POT Fraction: 80.1% good physics data
€ 380.1% due to bad beam data (at ND)
€ Generally smooth running
€ Replaced another 6V power supply
» Analog power supplies using
electrolytic capacitors
failing (15 years old)
» Team at Soudan can repair these
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